Oct 19 due Dec 2 HERE

Charges on a Sphere

xfpp3d

The Game of Life

xflat2d

Solitons

Solitons and Nonlinear Wave Equations

Solution of Nonlinear Systems using Newton’s Method: Summary Notes: (PS Problem handout (PS PHY/CAM 381C (1997, UT Austin) Problem set 6 (PS) and key (PS PHYS410 (2000) Problem set 5 (PS) and key (PS PHYS410 (2000)

xvs visualization server

Schrodinger Equation

PS PDF

xvs visualization server The demonstration programdemocomplex.f

Chaos

The Rings of Saturn

xfpp3d IRIS/NAG Explorer

Traffic Simulation using Cellular Automata

Moon Landing

2D Ising Model

xflat2d

Dissipative Gas Simulation

xflat2d

Particle Physics Simulations Reaction-Diffusion Equations Toda Lattices Neural Networks Genetic Algorithms Simulated Annealing Pedagogical Java Applet

Quantum Computation